Dear Mr Manuel E. Hernandez,

Thank you for reviewing our manuscript, “Changes in the body posture of women occurring with age", and for all of your valuable comments.

Based on your remarks, the following corrections were made:
1. Final conclusions were amended so they were conclusions and not results.
2. The sentences referred to in point 2 were removed, as they were unclear for the reader.
3. The methodology section specified the accuracy of photogrammetry and cited relevant studies on its reliability compared to radiological examination.
4. The discussion underlined the clinical importance of the obtained results.
5. The table caption was corrected, as suggested by the reviewer.
6. The manuscript was carefully reviewed from a linguistic point of view.

Should you have any further questions, please let us know so that we can improve our paper.

Yours sincerely,

The authors